Author's response to reviews

Title: The utility of clinical decision tools for diagnosing osteoporosis in postmenopausal women with rheumatoid arthritis.

Authors:

Caroline Brand (Caroline.Brand@mh.org.au)
Adrian Lowe (Adrian.Lowe@mh.org.au)
Stephen Hall (drshall@ozemail.com.au)

Version: 4 Date: 23 January 2008

Author's response to reviews:

The BioMed Central Editorial Production Team
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923

Dear Sir / Madam

Re: Formatting changes as requested

- Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. [Done - page 1]

- Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system [Done]

- Country - Please remember to include the country in the affiliation details, after the postcode. [Done - page 1]

- Please remove the key words. [Done]

- Please remove the word count from the abstract title and the top of the background. [Done]

- Please update the abstract so that it is identical to the abstract in the manuscript. [Done]

- Authors' contributions - Please indicate that all authors read and approved the final manuscript. [All authors have read and approved the final manuscript]

- References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used. [Done - pages, 14, 15 and 16]

- Table order - The first table referenced in the manuscript text should be Table 1, the second, Table 2, etc. The tables should be provided in this same order at
the end of the manuscript. [Done ¿ pages 18, 19 and 20]

- Every table provided must be referred to in the text. [Done - page 8 second paragraph]

- References - References should be cited in the text using consecutive numbers in square brackets. We notice the references are out of order. Please correct this. [Done]

- Some revisions have been made in Table 2. 18 and 12 have been added (in the 3rd and 4th rows Femoral Neck and Lumbar spine), (3rd column T< -2.5). [Page 19]

Yours sincerely
Caroline Brand